Author's response to reviews

Title: The involvement of AMPK/GSK3-beta signals in the control of metastasis and proliferation in hepato-carcinoma cells treated with anthocyanins extracted from Korea wild berry Meoru

Authors:

Song Yi Park (songyip3507@gmail.com)
Yun-Kyoung Lee (yunkyoung0225@gmail.com)
Won Sup Lee (lwshmo@hanmail.net)
Ock Jin Park (ojpark@hnu.kr)
Young-Min Kim (kym@hnu.kr)

Version: 8 Date: 25 February 2014

Author's response to reviews: see over
February 25, 2014

Mr Tom Rowles, Senior Executive Editor
BMC complementary and alternative medicine

Dear Tom,

It is with excitement that I resubmit to you a revised version of manuscript 5718954999889965, Dyadic Perceptions of Goals, Conflict Strategies, and Perceived Resolvability in Serial Arguments for the BMC complementary and alternative medicine. Thank you for giving me the opportunity to revise and resubmit this manuscript. In keeping with my last communication with you, I am resubmitting this revision before the agreed upon deadline, 25 February 2014. We have addressed the editor’s concerns by revising the manuscript. The point-by-point replies are given in this letter. We hope that we have addressed satisfactorily all concerns raised by the reviewer, and that this manuscript is now suitable for publication.

Sincerely yours,

Young Min Kim, Ph.D.
Hannam University
Email: kym@hnu.kr

<Response to editor>

Comment

1. Specifically, rather than information on the isolation of anthocyanins used in your study, we require further information on the Meoru fruit used. Specifically, we would ask you to confirm who formally identified the collected fruit to ensure that the correct plant material was being used. In order to comply with our editorial policies (http://www.biomedcentral.com/about/editorialpolicies#Ethics), we also require that a voucher specimen of this plant material be deposited in a publicly available herbarium for
future reference. This fact should be detailed in your manuscript, alongside a voucher number for the sample.

2. Please also ensure that your revised manuscript conforms to the journal style (http://www.biomedcentral.com/info/ifora/medicine_journals). It is important that your files are correctly formatted.

Response

1. We added the voucher specimen of Vitis coignetiae Pulliat (known as meoru in Korea). We added the manuscript on the materials and methods on page 5, line 17-18. The plant with voucher specimen number KNKA200506031111 was deposited in the Korea national arboretum.

2. We revised manuscript conforms to the journal style.